Author's response to reviews

Title: Clinical spectrum of rhabdomyolysis presented to Pediatric Emergency Department

Authors:

- Chun-Yu Chen (116984@cch.org.tw)
- Lu-Lu Zhao (zhaolulu2001@tzuchi.com.tw)
- Wen-Chieh Yang (v1004@hotmail.com)
- Yu-Jun Chang (83686@cch.org.tw)
- Kang-Hsi Wu (d5284@mail.cmuh.org.tw)
- Han-Ping Wu (arthur1226@gmail.com)

Version: 2 Date: 19 January 2013

Author's response to reviews: see over
Jan. 19th, 2013

Dear Editor-in-Chief,

Enclosed you will find the revised manuscript entitled "Clinical Spectrum of Rhabdomyolysis Presented to Pediatric Emergency Department". In this revised manuscript, we have addressed your concerns and made the mandatory style changes to meet the style of the journal: BMC Pediatrics. We wish to thank you for allowing us to resubmit this revised manuscript and hope that you will now find it acceptable for publication in the BMC Pediatrics. I would very like to make any further revision followed by your comments and it would be highly appreciated if this manuscript could be smoothly run to publication process.

Best regards,

Han-Ping Wu, MD, PhD
Director, Department of Pediatrics
Buddhist Tzu-Chi General Hospital, Taichung Branch
Tel: +886-4-36060666 ext 3117
Fax: +886-4-36021123